Author’s response to reviews

Title: 2-methylbutyryl-CoA dehydrogenase deficiency associated with autism and mental retardation: a case report

Authors:

Oivind J Kanavin (oivind.kanavin@ullevaal.no)
Berit Woldseth (berit.woldseth@rikshospitalet.no)
Egil Jellum (egil.jellum@medisin.uio.no)
Bjorn Tvedt (bjorn.tvedt@ullevaal.no)
Brage Andresen (brage@au.dk)
Petter Stromme (petter.stromme@medisin.uio.no)

Version: 2 Date: 6 June 2007

Author’s response to reviews:

This case report has been reformatted in order to comply with BMC Medical Case Reports. As noted in email June 4 by assistant editor Claire Shaw we have shortened the reference list to include 15 references. We have also modified the title which ends with "a case report". The mother's consent is added under acknowledgements.